Author's response to reviews

Title: Primary pineal melanoma presenting with leptomeningeal spreading in a 22-year-old woman: a case report

Authors:

Parisa Azimi (parisa.azimi@gmail.com)
Hassan Reza Mohmmadi (mohammadi5@sbmu.ac.ir)
Mitra Refiezadeh (m.refiezadeh@yahoo.com)

Version: 4 Date: 7 February 2012

Author's response to reviews:

The Journal of Medical Case Reports Editorial Team,
MS: 6829398065755325
'Primary pineal melanoma presenting with leptomeningeal spreading in a 22-year-old woman: a case report'
The draft was reviewed and the style and vocabulary was corrected as suggested.